Author’s response to reviews

Title: 10-year ASCVD risk is positively correlated with depressive symptoms in a large general population

Authors:

Guozhe Sun (gzhsun66@163.com)
Ning Ye (yening_cmu@163.com)
Shaojun Wu (1145167664@qq.com)
Ying Zhou (zhouying8111003@126.com)
Yingxian Sun (cmu1h_syx@126.com)

Version: 1 Date: 19 Dec 2018

Author’s response to reviews:

Dear Dr. Peter Dome,

We have read through the manuscript and revised some errors and typos. For the 'Availability of data and materials' section, we stated that data would not be shared, and provided the reasons in the manuscript text.

Thank you very much for your suggestion and consideration.